Author's response to reviews

Title: miR-125b induces cellular senescence in malignant melanoma

Authors:

Anne Marie Rie Nyholm (rienyholm@gmail.com)
Catharina M Lerche (cler0001@bbh.regionh.dk)
Valentina Manfe (valentinamanfe@libero.it)
Edyta Biskup (edyta.biskup@gmail.com)
Peter Johansen (pjoh0809@gmail.com)
Niels Morling (niels.morling@forensic.ku.dk)
Martin Glud (glud.martin@gmail.com)
Robert Gniadecki (r.gniadecki@gmail.com)

Version: 2 Date: 8 August 2013

Author's response to reviews:

To The Editor

Please, find enclosed a revised version of the manuscript. As requested, we have provided a sentence with a reference to the compliance with the national guidelines for animal experimentation.

With kindest regards,
on behalf of all authors,

Robert Gniadecki, prof. dr. med.